

### Are group members less inequality averse than individual decision makers?

Haoran He, Marie Claire Villeval

### ▶ To cite this version:

Haoran He, Marie Claire Villeval. Are group members less inequality averse than individual decision makers?. Journal of Economic Behavior and Organization, 2017, 138, pp. 111-124. halshs-00996545v4

### HAL Id: halshs-00996545 https://shs.hal.science/halshs-00996545v4

Submitted on 4 Apr 2017

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

### Forthcoming in Journal of Economic Behavior & Organization

# Aregroup members less inequality averse than individual decision makers?

Haoran He and Marie Claire Villeval

### April 2017

**Abstract**:Do groups exhibit more or less inequality aversion than individuals? Although the previous literature has shown that in many environments individuals in groups make more selfish decisions than when deciding in isolation, we findthat individuals express *more* inequality aversion when making initial proposals in a group decision-making environment compared to an individual decision-making environment. This may bedriven by a change in the decision-making environment and by beliefs about the prevailing norm in the group, but we exclude that it is driven by a loss of anonymity or by efficiency concerns. By investigating how groups aggregate individual preferences under a unanimity rule, we show that the members with median social preferences lead thegroup decisions and a higher inequality aversion compared to the median slows down the convergence process. Overall, final decisions in groups reveal the same level of inequality aversion than individual decisions.

**Keywords:**Group, inequality aversion, preference aggregation, social image, experiment **JEL classification:** C91, C92, D03, D63, D72

#### **Contact Information**:

Haoran He, School of Economics and Business Administration, Beijing Normal University, 19, XinJieKouWai Street, HaiDian District, Beijing 100875, P. R. China. E-mail:haoran.he@bnu.edu.cn.

Marie Claire Villeval, Univ Lyon, CNRS - GATE, 93, Chemin des Mouilles, F-69130 Ecully, France; IZA, Bonn, Germany; University of Innsbruck. E-mail: villeval@gate.cnrs.fr. Tel.: +33 472 86 60 79.

**Acknowledgments**: We are grateful to L. Balafoutas, F. Galeotti, M. Joffily and participants at the Asia-Pacific Meeting of the Economic Science Association in Auckland and the ASFEE conference in Besançon for their useful comments. We thank Y. Yang, S. Onderstal and A. Schram for sharing the software of their production game with us. We thank W. Wang and P. Peifor excellent research assistance. This research program has been supported by a grant from the LABEX CORTEX (ANR-11-LABX-0042) of Université de Lyon, France, within the program "Investissements d'Avenir" (ANR-11-IDEX-007) operated by the French National Research Agency (ANR), and a grant from the National Natural Science Foundation of China (Proj. No. 71303022).

#### 1. INTRODUCTION

Social comparisons, both among individuals and among groups, are widespread in human societies. Whilesome individuals enjoy outperforming others, many peopleare inequality averse.In economic models such as Fehr and Schmidt (1999) and Bolton and Ockenfels (2000), inequality aversion captures the fact that people care about both their own material payoff and the distribution of payoffs between them and others. To date, the experimental literature has almost exclusively considered inequality aversion when an individual interacts with other individuals. It has less deeply investigated inequality aversion when individuals decide as members of a group, although the norm of equality in groups has been shown to be often more appealing than the norm of efficiency when groups are heterogeneous (Nikiforakis et al. 2012; Reuben and Riedl, 2013; Gangadharan et al., 2015). Social dynamics, in particular the influence of peers, may generate systematic differences in preferences compared to an environment in which people decide in isolation. It is unclear, however, whether inequality aversion is stronger or weaker in a social environment than when individuals interact with a single other individual. We know even less about inequality aversion when groups interact with other groups. Group members may weigh less the difference with another group; on the opposite, they may behave more competitively than when interacting with a single person, expressing more disadvantageous and less advantageous inequality aversion because of the influence of group identityon behavior (Akerlof and Kranton, 2000; Chen and Li, 2009).

In this paper, we designed a laboratory experiment to compare inequality aversion in individuals when these individuals interact with another person and when they interact as a member of a groupfacing with another group, using various allocation tasks. Weaddressthree questions. First, we investigate whether the degree of inequality aversion when group members

make initial proposals to the group differs from whenindividual choices are made in isolation (*i.e.*, when interacting with a single individual). Studying the first proposal made by individuals in a group decision-making environment before they receive any feedback about others' choices (and not only the final group decision, as done usually in the literature) allows us to isolate the impact of a collective decision-making context independently of the social information conveyed by peers' proposals in the next rounds. When making their first proposal to the group, individuals may express less inequality aversion than when making decisions in isolation if a group environment encourages selfishness; on the opposite, they may express more inequality aversion to compensate for the expected selfishness of others.

Second, we study the process of aggregation of individual proposals to form the final group decision and we examine whether it varies with the degree of inequality aversion that was expressed initially. Byobservingthe members'initial proposals and measuring the distance with the final votewhile keeping the group environment constant, we can characterize the formation of group decisions once people learn about others' proposals. Analyzingthe entire dynamics of the group decision formationallows us to address the question of who in the group, in terms of relative inequality aversion, has a stronger influence on the final decisions.

Finally, we study whether individual preferences in a group decision-makingenvironment depend on whether the anonymity of group members is preserved or not during the aggregation process, revealing the possible role of social image concerns.

We contribute to the literature comparing group and individual decision-making. Many studies have found that groups behavein general more rationally and selfishly than individuals (Charness and Sutter, 2012), although somehave shownthat the differencedepends crucially on the nature of the task and on the decision-making procedure (*e.g.*, Kocher and Sutter, 2007).

However, these studies did not explore inequality aversion directly. A recent exception is Balafoutas *et al.* (2014)who show that while groups express the sameadvantageous inequality aversionas individuals, they are more benevolent than individuals in the domain of disadvantageous inequality and much more efficiency-oriented. Our design introduces three main differences. We measure inequality aversion under the Fehr and Schmidt's (1999) theoretical framework. We use games in which the fixed option maximizes the level of inequality, instead of fixing payoff equality. And we isolate image concerns in the aggregation of preferences.

We also contribute to the literature by matching this comparative analysis of individual and group decision-makingregarding inequality aversion with the analysis on the aggregation of individual preferences in groups (*e.g.* Gillet *et al.*, 2009; Zhang and Casari, 2012; Ambrus *et al.*, 2014). Bycomparing the individual choicesmade in isolation before any social interaction and those made in a group environment, we can explore whether some players have a stronger influence in the group decision-making process. In particular, we test the hypothesis that those with a median level of inequality aversion make less concessions than other group members, although all players have a veto power under the unanimity rule introduced in our experimental design.

Another contribution is related to the study of whether and how the anonymity of decisions affects individual initial proposals in a group environment and their adjustment during the aggregation process. In real settings, choices by voters in various policy-making procedures are typically anonymous, while choices by juries, boards, and families usually result from non-anonymous interactions. When it is common information that a proposal emanates from a physically identified group member, allocation choices may express a different degree of inequality aversion than when choices are made anonymously. Indeed, previous literature has

shown that individuals tend to make more selfless decisions when observed because they care about their social image (*e.g.*, Andreoni and Petrie, 2004; Soetevent, 2005, 2011; Benabou and Tirole, 2006; Ariely *et al.*, 2009; Linardi and McConnell, 2011; Reinstein and Riener, 2012; Karlan and McConnell, 2012). We study whether a similar effect is observed in the context of our experiment.

Precisely, in our experimentwe elicit advantageous and disadvantageous inequality aversion at the individual level by means of the multiple price listsintroduced by Blanco *et al.* (2011), based on the Ultimatum Bargaining Game (Güth *et al.*, 1982) and a Modified Dictator Game (originally developed by Forsythe *et al.*, 1994; Hoffman *et al.*, 1994). Weadapt this design to a groupdecision-making environmentwhenall members of a groupreceive the same payoff from the group decision. Pairs of three-player groups perform the same allocation tasks. The group decisions result from votes made undera unanimity rule. Using both within-subject and betweensubject designs allows us to compare individuals' decisions made in isolation and their initial and final proposals in a group environment. To identify the role of anonymity, we make a betweensubject comparison withanadditional treatment in which subjectscan physically identify their group members and their proposals.

We have threemainfindings based on the analysis of switching points in the two games. First, on average individuals express more disadvantageous and advantageous inequality aversion when they make their initial proposals to the group than when they decide in isolation. This increased inequality aversion is driven neither by social image concerns, as the lift of anonymity has little effect, nor by efficiency concerns, as similar differences observed in both games although efficiency is kept constant only in the Ultimatum game, nor by peer effects since no social information has been disseminated yet. It may result from the expectations about the social norm

prevailing in the group orfrom a change in preferences due to a different decision-making environment. Second, the degree of inequality aversion revealed by the final proposals, which is also the group decisions, is similar to that observed when individual decisions are made in isolation, which indicates the importance of social information on the evolution of individual proposals. Third, we show that the group members with the median level of inequality aversion drive the aggregation process. A higher degree of inequality aversion compared to the median slows down the convergence process.

The remainder of this paper is organized as follows. Section 2 briefly reviews the related literature. Section 3 presents the experimental design and procedures. Section 4 analyzes the results, and Section 5 discusses these results and concludes the paper.

#### 2. RELATED LITERATURE

Our main contribution is to connect literatures on inequality aversion and on group decisionmaking. Tests of inequality aversion models havefirst been developed at the aggregate level (Fehr and Schmidt, 1999; Bolton and Ockenfels, 2000). Then, studies attempted toelicitthese preferencesat the individual level(Engelmann and Strobel, 2004; Bolton and Ockenfels, 2006; Dannenberg *et al.*, 2007; Güth *et al.*, 2009; Bartling *et al.*, 2009; Blanco *et al.*, 2011; Beranek *et al.*, 2015; Yang *et al.*, 2016). Within-subject testshave producedmixed evidence. Engelmann and Strobel (2004) find no support for eitherFehr and Schmidt's or Bolton and Ockenfels' modelsin a simple distribution game. Blanco *et al.* (2011) conclude that the predictive power of Fehr and Schmidt's model is limited at the individual level. In contrast, Dannenberg *et al.* (2007) show that in social dilemmas thedisadvantageous inequality aversionparameter has some explanatory power.These testshave used sequential prisoner's dilemma games (Blanco *et al.*, 2011)or public goods games (Blanco *et al.*, 2011;Dannenberg *et al.*, 2007). Using different gamesYang *et al.* 

(2016)show the robustness of the model to efficiency concerns and variations in payoff scales.In our paper weadjustthe games used in Blanco *et al.* (2011) to agroupdecision-making environment.

As regards the preferences expressed by groups, on the one hand studies have shown the strength of the norm of equality (Nikiforakis *et al.* 2012; Reuben and Riedl, 2013; Gangadharan *et al.*, 2015). On the other hand,many have found that groupsbehave more rationally in non-strategic interactions<sup>1</sup> and more selfishly than individuals in manygames(Kugler *et al.*, 2012),<sup>2</sup>although not all.<sup>3</sup> Thesemixedfindings may be due to several factors, namely different preferences in groups than in individual interactions driven by a different context or by strategic concerns, the skewness of the distribution of individual members' preferences, orthe aggregation process. While Gillet *et al.* (2009) have shown that under the majority rule the median voter departs from his individual preferences after observing more selfish proposalsthan her own preferences, neutralize each other (thus if the median members' pro-social preference is belowthemean, it drives the group toward greater selfishness).However, the comparison of inequality aversion in individual and in group interactions has remained almost unexplored.<sup>4</sup>

<sup>1</sup>Groups make fewer mistakes (Fahr and Irlenbusch, 2011), suffer less from hindsight bias (Stahlberg *et al.*, 1995), myopic loss aversion (Sutter, 2007), and overconfidence (Sniezek, 1992). Results on risk attitudes are less consistent: some found thatgroups are more risk averse (Baker *et al.*, 2008; Shupp and Williams, 2008) but others have shown that groups are closer to risk neutrality (He *et al.*, 2012) or that they take better risks (Rockenbach *et al.*, 2007). <sup>2</sup> This result holds for dictator games (Luhan *et al.*, 2009), sequential games such as ultimatum (Robert and Carnevale, 1997; Bornstein and Yaniv, 1998), trust (Cox, 2002; Kugler *et al.*, 2007; Song, 2009), centipede (Bornstein *et al.*, 2004a), power-to-take games (Bosman *et al.*, 2006) and signaling games (Cooper and Kagel, 2005), as well as simultaneous games such as public goods (Van Vugt *et al.*, 2007; Gillet *et al.*, 2009), beauty contests (Kocher and Sutter, 2005; Kocher *et al.*, 2006; Sutter, 2005), and auctions (Cox and Hayne, 2006; Sutter *et al.*, 2009; Sheremeta and Zhang, 2010; Casari *et al.*, 2011; Cheung and Palan, 2011).

<sup>&</sup>lt;sup>3</sup> Cason and Mui (1997) find in a dictator game that teams that are initially more self-regarding tend to act less selfishly. This polarization is due more to social comparisons (which give more weight to pro-social individuals) than to persuasion. In a similar game, Franzen and Pointner (2014) observe no difference. Müller and Tan (2013) find less selfish groupchoices in sequential market games, but mixed evidence is found in gift-exchange games (Kocher and Sutter, 2007).

<sup>&</sup>lt;sup>4</sup>Note that one can find studies on how inequality aversion in groups affects contractual design (Rey-Biel, 2008; Bartling and von Siemens, 2010), sharing rules (Gill and Stone, 2015), peer pressure (Mohnen *et al.*, 2008), or sanction and cooperation (Masclet and Villeval, 2008).

As mentioned earlier, one exception is Balafoutas et al. (2014)whostudy distributional preferences under both individual and teamregimes. They find that teamseliminate choices consistent with inequality aversion and spitefulness and they favor efficiency, in particular because communication allows efficiency-loversto be more assertive than others. While our objective is also to better understand how social preferences are aggregated in groups, we differ from this study in several respects. Balafoutas et al. (2014) elicit distributional preferences based on the double-price list technique of Kerschbamer (2015): subjects make binary allocation choices, one choice involving always the same equal payoff and the other choice asymmetric payoffs; in half of the decisions, the decision-maker is ahead the passive agent and in the other half he is behind. In contrast, we use the UG and the MDG of Blanco et al. (2011) and in the latter the fixed option maximizes the level of inequality. It is interesting to compare whether the conclusions are similar when the fixed option is equality (Balafoutas et al., 2014) or the highest inequality (our design). Moreover, Balafoutas et al. (2014) treatteamsas a decision unit; instead, we consider group members as individuals and characterize how a group environment affects each member's individual preferences. In addition, our design allows us to isolate the role of anonymity in groups. Another contribution of our approach is that we combine a between-subject design and a within-subject design to measure whether different people and and/or the same people express different preferences in individual and group decision-making environments.<sup>5</sup>

The combination of the decision-making procedure and the distribution of players' types may determine the differences between individual and group decision-making (seeBosman *et al.*,

<sup>&</sup>lt;sup>5</sup> There are other differences with our design. Contrary to Balafoutas *et al.*, our games are played in a single session (instead in two consecutive weeks), we alternate the order between group and individual decisions, and we do not allow free communication. In their design, unanimity must be reached in five rounds maximum, while in our case we apply a time constraint. They pay subjects both as an active player and as a passive person while we pay subjects randomly in one of the two roles. In case of inability to reach an agreement, their subjects' payoffs are null, which may create a stronger pressure to reach unanimous group decisions than in our case where group decisions are randomly assigned from the possible decisions.

2006). Previous literature has shown that groups do notreach the same decisions as individuals when a majority is sufficient to make a decision or when unanimity is required (*e.g.* Bornstein *et al.*, 2004b; Blinder and Morgan, 2005; Gillet *et al.*, 2009).<sup>6</sup> Moreover, the mode of communication may matter. Manystudies use a face-to-face protocol or unrestricted communication via chat boxes(Kocher and Sutter, 2005; Kocher *et al.*, 2006; Kugler *et al.*, 2007; Sutter *et al.*, 2009; Ambrus *et al.*, 2014;Balafoutas *et al.*,2014). Kocher and Sutter (2007) show that groups behave more selfishly than individuals in an anonymous computerized procedure but not in a face-to-face protocol.Furthermore, anonymitymay affect the process of deindividuation within groups.<sup>7</sup>In our paper, we impose unanimity and restricted communication instead of face-to-face. This allows us to identify the role of anonymity while keeping the environment constant.

#### **3. EXPERIMENTAL DESIGN AND PROCEDURES**

In this section, we first introduce the games, then the treatments and the matching design, and finally, the experimental procedures.

#### 3.1.The games

We first describe the individual decision-making environment before introducing the group decision-making environment.

#### Inequality aversion in an individual decision-making environment

To estimate the individuals' disadvantageous and advantageous inequality aversion, we replicate two of the games used in Blanco *et al.* (2011). Eachgame consists of 21 decision

<sup>&</sup>lt;sup>6</sup> Many papers on group decisions impose unanimity (*e.g.*, Sutter, 2005; Kocher and Sutter, 2005, 2007; Shupp and Williams, 2008; Luhan *et al.*, 2009; Sutter *et al.*, 2009). Some use the majority (Baker *et al.*, 2008; Harrison *et al.*, 2012) or the median (Bischoff and Krauskopf, 2013). Others allow for unrestricted deliberation (Cason and Mui, 1997; Bornstein *et al.*, 2004a; Bosman *et al.*, 2006; Schupp and Williams, 2008; Ambrus *et al.*, 2014).

<sup>&</sup>lt;sup>7</sup> Anonymity is a key factor of deindividuation. In social psychology, the deindividuation theory of Festinger *et al.* (1952) predicts that the anonymity of individuals in a group may lower their sense of personal identity and reduce compliance with the group norm. In contrast, the social identity model of deindividuation (Reicher *et al.*, 1995) suggests that anonymity facilitates the alignment of the individual with the group's preferences.

problems, as shown in Table 1.The games are played under the veil of ignorance using the strategy method.<sup>8</sup>

The Ultimatum Game (UG, hereafter) involves a proposer and a responder. The proposer must share a pie of 400 points between himself and the responder. He makes an offerSto the responder, keeping (400 - S) to himself. If the responder rejects the offer (option A), both players earn zero. If the responder accepts the offer (option B), the share is implemented. The proposers' offers are restricted to multiples of 20, leading to 21 distributions from (400, 0), (380, 20), ... to (0, 400). Subjects make their choices in each of the two roles sequentially on two separate screens to minimize interactions between the two decisions.

In the Modified Dictator Game (MDG, hereafter), the dictator decides how many of 400 points she is willing to sacrifice to equalize payoffs between herself and the receiver. There are 21 decision problems with two options. The left option always pays 400 points to the dictator and nothing to the receiver. The right option gives equal payoffs to both players and varies from (0, 0), (20, 20), ... to (400, 400). Each subject makes a choice in the role of a dictator.

	Ulti	matum Game	Modi	fied Dictator Game		
Decision	Proposer's	Responder	's decision	Di	ctator's decision	
problem	decision	Option A	Option B	Option	A Option B	
1	(400, 0)	Reject	Accept	(400,	0) (0, 0)	
2	(380, 20)	Reject	Accept	(400,	0) (20, 20)	
3	(360, 40)	Reject	Accept	(400,	0) (40, 40)	
4	(340, 60)	Reject	Accept	(400,	0) (60, 60)	
5	(320, 80)	Reject	Accept	(400,	0) (80, 80)	
6	(300, 100)	Reject	Accept	(400,	0) (100, 100)	
7	(280, 120)	Reject	Accept	(400,	0) (120, 120)	
8	(260, 140)	Reject	Accept	(400,	0) (140, 140)	
9	(240, 160)	Reject	Accept	(400,	0) (160, 160)	
10	(220, 180)	Reject	Accept	(400,	0) (180, 180)	
11	(200, 200)	Reject	Accept	(400,	0) (200, 200)	
12	(180, 220)	Reject	Accept	(400,	0) (220, 220)	
13	(160, 240)	Reject	Accept	(400,	0) (240, 240)	

Table 1. The Ultimatum Game and the Modified Dictator Game

<sup>8</sup> Brandts and Charness (2011) survey the literature to compare the strategy method and the direct-response method. A total of 16 out of the 29 comparisons show no difference, four find differences and nine find mixed evidence.

14	(140, 260)	Reject	Accept	(400, 0)	(260, 260)
15	(120, 280)	Reject	Accept	(400, 0)	(280, 280)
16	(100, 300)	Reject	Accept	(400, 0)	(300, 300)
17	(80, 320)	Reject	Accept	(400, 0)	(320, 320)
18	(60, 340)	Reject	Accept	(400, 0)	(340, 340)
19	(40, 360)	Reject	Accept	(400, 0)	(360, 360)
20	(20, 380)	Reject	Accept	(400, 0)	(380, 380)
21	(0, 400)	Reject	Accept	(400, 0)	(400, 400)

Note: The first numbers in parentheses display the proposer's payoffs, the second numbers the receivers' payoffs.

In both games, we impose the restriction of single switching between the two options in the 21 problems.<sup>9</sup> Specifically, in the UG responders choose the number of the decision problem from which they accept all of the proposer's offers; in the MDG dictators select the number of the decision problem from which they always choose equal sharing. It was made clear that the subjects could switch from the first problem and that they were allowed not to switch at all. This gives each responder in the UG a single minimum acceptable offer that determines hisdegree of disadvantageous inequalityaversion. In the MDG the maximum amount that the dictator is willing to sacrifice to implement equal sharing determineshisdegree of advantageous inequalityaversion. Random draws at the end of the session determined the actual role in each game and which one of the decisions in each game was paid.

Inequality aversion in agroup decision-making environment

In a group environment, we pairedgroups of three subjects who play a collective version of the previously described UG and MDG. We use the same tables as for decisions made in isolation. To hold themonetary incentives comparable across individual and group conditions, the payoffs achieved in the group games are paid to each group member. For example, if the selected decision

<sup>&</sup>lt;sup>9</sup>Based on basic rationality axioms, rational players with monotone preferences should switch only once from option A to option B because their payoff becomes larger in the UG for all problems beyond the switching point; similarly in the MDG, the egalitarian outcome is always cheaper beyond this point. The same procedure has been applied notably by Tanaka *et al.* (2010) to elicit risk preferences and time consistency. We acknowledge that enforcing exactly one switching point may bias the choices of individuals who in the UG would like to reject splits giving them less than the equal share and those giving them more than the equal share. Imposing single switching, however, rules out inconsistent choices and more importantly, facilitates group decision-making by simplifying the aggregation process (although no theoretical foundation implies that the aggregation of rational individual preferences in a group should result in a single switching point), as explained below.

in the DG pays 400 points to the dictator group and leaves nothing for the receiver group, each of the three dictator groupmembers earns 400 points and each of the three receiver group membersreceives 0.The actual roles of agroupare randomly assigned at the end of the session and one decision problem in each of the UG and the MDG is randomly selected for payment.

Unanimity is required to form a groupdecision. Choosing unanimity instead of the majority rule allows us to study the convergence process to the group decisionmore clearly under the circumstance thateach player is given a veto power. Specifically, in each game the groupmembers mustsimultaneously submit their individual proposal for the group decision. Then the three proposals are displayed on themembers' screens. If theyare not identical, a new round starts and each member mustsubmit a new proposal (possibly the same as in the previous round). This procedure is repeated until all groupmemberssubmit identical proposals. The number of rounds is unrestricted within the limit of 10 minutes for each groupdecision. In case unanimityhas notbeen reachedafter the 10 minutes have elapsed, the programselectsrandomly one of all possible decisions. We have preferred a formal process of decision-making with restricted communication instead of using face-to-face to have a better control of the interactions within groupsand to be able to isolate in a separate treatment the role of anonymity while keeping the rest of the environment constant. Thisalso allows us to observe the evolution of proposals, as all members make exactly the same number of proposals.

One advantage of our design is that for each subject, we are able to observe his individual decision made in isolation, his initial proposal in the group before learning others' preferences, and his final decision as aggregated in the group decision.

#### 3.2. Treatments and matching protocol

The experiment consists of three main treatments using a between-subject design. Each treatment includes four mainparts that allow us to make within-subject comparisons across parts.<sup>10</sup>Parts 1 and 2 correspond to the one-shot UG and MDG played individually, whereasParts 3 and 4 differ across treatments. The I-I(for Individual-Individual) treatment involves only individual decisionmaking: Parts 3 and 4 replicate Parts 1 and 2 (UG and MDG). TheI-AG (for Individual-Anonymous Group)treatmentintroduces collectivedecision-making in Parts 3 and 4 (UG and MDG). Group members do not know whom they are interacting with in their groups. The I-NAG(for Individual-NonAnonymous Group)treatment is identical to the I-AGtreatmentwith two exceptions. First, welift anonymity within the group: at the beginning of Part 3, before they make their decisions, members are told that the three subjectsseated in the same row belong to the same group, with identification numbers I, II, and III assigned to the members seated at the left, middle and right of the row, respectively.<sup>11</sup>Second, theidentification number of each member appears next to her/his proposals so that group memberscan trace the evolution of all the three members' proposals across rounds. Lifting anonymity may expose subjects to a higher social pressure, which may influence both their degree of inequality aversion and their bargaining behavior. In contrast, players receive no information on the composition of the group they are paired with and on the proposals made within theother group. To control for possible order effects, we conducted an additional NAG-I treatment. Compared totheI-NAG treatment, the appearance order of Parts 3 and 4 and Parts 1 and 2 is reversed. This allows us tostudywhether decisions made in isolation after group decisions differ from those made before group bargaining.

<sup>&</sup>lt;sup>10</sup>In fact, in a fifth part we replicated the production game of Yang *et al.* (2016) in the original individual version or in a novel group decision-making environment to study whether inequality aversion as measured in our main games correlates with behavior in this game. For saving space, we do not report the details of this game and its results in this paper but they can be found in our working paper (He and Villeval, 2015). Inequality aversion, as measured in our games, is not correlated with behavior in the production game.

<sup>&</sup>lt;sup>11</sup> Although subjects are seated in cubicles, they can move their seat a bit and see the face of their neighbors without having to stand up. They are also able to talk to their group members at the end of the session.

In all treatments, the appearance order of the UG and the MDG was randomized across sessions, but the order of the two games in the same session was held constant in Parts 1 and 2 and in Parts 3 and 4. A perfect stranger matching protocol rules out reciprocity and reputation building across parts. Each group(individual) is paired with a different group(individual) across parts, whereas the composition of each groupis kept constant across parts. Subjects are informed in each part that at the end of the experiment, the program will randomly pair them with another participant (group) in the room and will randomly assign the two roles.

Table 2 summarizes the key features of our experimental design.

Treatment	Part 1	Part 2	Part 3	Part 4	Part 5
I-I	Individual	Individual	Individual	Individual	Individual
	UG/MDG	MDG/UG	UG/MDG	MDG/UG	PG
I-AG	Individual	Individual	GroupUG/MDG	GroupMDG/UG	GroupPG
	UG/MDG	MDG/UG	Anonymity	Anonymity	Anonymity
I-NAG	Individual	Individual	GroupUG/MDG	GroupMDG/UG	GroupPG
	UG/MDG	MDG/UG	No anonymity	No anonymity	No anonymity
NAG-I	GroupUG/MDG	GroupMDG/UG	Individual	Individual	GroupPG
	No anonymity	No anonymity	UG/MDG	MDG/UG	No anonymity

Table 2. Summary of the experimental design

*Note*: UG for Ultimatum Game, MDG for Modified Dictator Game, and PG for Production Game (not analyzed in this paper).

#### 3.3. Procedures

The experiment was conducted at the laboratory of Beijing Normal University. 336 volunteers were recruited via announcements on the bulletin board system andin accommodation and teaching buildings of local universities. Each of the 14 sessions involved 24 subjects (2 sessions with I-I and 4 with each other treatment). In total, we have48 individual observations for the I-I treatment and 32 group observations for each other treatment. Due to inability to reach unanimityfor the final group decisions, we losta few observations in the role of the dictator in the MDG(1 in bothI-AGand NAG-I) and in the role of the proposerin the UG(4 in I-AGand 2 in NAG-I).

The experiment was computerized using z-Tree (Fischbacher, 2007). Upon arrival, the subjects were assigned randomly to a computer terminal. Each part was introduced sequentially after completion of the previous one. Instructions were distributed and questions were answered in private (see Appendix 1). Subjects were given no information about the number of parts and they received no feedback on the outcomes of any part until the end of the experiment.Sessions lasted approximately 90 minutes. Subjects received in cash the sum of their earnings for all parts from an assistant who was unaware of the content of the experiment. This was made common informationin the instructions. In the experiment, we used a conversion rate of 100 points= 3 Yuan≅US\$ 0.84 in 2015 PPP. Participants earned on average 82.70 Yuan (about US\$23.24), including a 10-Yuan show-up fee, which is above the average salary for a student's part-time job.

#### 4. RESULTS

In this section, we first analyze the subjects' aversion to disadvantageous inequalityand to advantageous inequalityas expressed in their individual decisions made in isolation and in their final decisions made in the group decision-making environment. Second, we study the process according to which individuals change their proposals progressively when they are in a group environment.

#### 4.1. Inequality aversionin individual and group decision-making environments

In this sub-section, we analyze individuals' disadvantageous and advantageous inequality aversion in both individual and group decision-making environments. For that purpose, we conduct our analysis using the values of the switching points in each game.Note that we do not use the estimated values of the  $\alpha$  and  $\beta$  parameters of the Fehr and Schmidt's (1999) model. This is for two reasons. First, this model has been designed for capturing individual behavior in an

individual decision-making environment and not in a group decision-makingenvironment. So, using these parameters for the decisions made in the group decision-making environment would not be meaningful. Second, estimating these parameters in the group decision-making environment would require assuming that in such an environment, the in-group members and the out-group members have the same weight in the individual's utility function. The literature on group identity has shown that this is frequently not the case (Akerlof and Kranton, 2000; Chen and Li, 2009).<sup>12</sup>

Table 3 reports the mean values of the switching pointin the various games for the individual decisions, the initial individual proposals and the group decisions, by treatment. The mean switching point indicated by the chosen decision number in the UG provides a measure of the acceptance threshold that we take as a proxy of disadvantageous inequality aversion. The mean switching point indicated by the chosen decision number in the MDG provides a measure of advantageous inequality aversion.

We first compare the degrees of inequality aversion revealed by the decisions in the individual decision-making environment and by the initial proposals in the group decision-making environment, before analyzing the groups' final decisions.

	Group en	Number of	
Individual environment	Individual initial proposals	Group decisions	subjects

Table 3. Mean switching points in the UG and the MDG in the individual and the groupdecision-making environments

<sup>12</sup>Nevertheless, we studied how the values of the  $\alpha$  and  $\beta$  parameters our individual decision-making environment compare with that found in the previous literature. We calculated these parameters as in Blanco *et al.* (2011), using non-linear monotonic conversion and income comparisons between the player and his co-participant(see details in Appendix 2).Using point estimates, Mann-Whitney rank-sum tests indicate no significant difference between the values of  $\alpha$  and  $\beta$  in our experiment and those reported in Blanco *et al.*Table A1 in Appendix 2 displays the distribution of the two parameters using the same intervals as Fehr and Schmidt (1999) and Blanco *et al.* (2011).Perhaps surprisingly, considering the differences in the cultural and political backgroundsbetween China and the U.K., the distribution of each parameter is similar in our experiment and in Blanco *et al.* or in Fehr and Schmidt.

	Mean	S.D.	Mean	S.D.	Mean	S.D.		
Switching point in the UG (acceptance threshold)								
I-I	6.00	3.50	-	-	-	-	48	
I-AG	6.43	3.99	7.23***	3.73	7.04	3.04	84	
I-NAG	6.14	3.30	7.09****	3.35	7.03**	3.06	96	
NAG-I	6.02	3.46	6.76***	3.66	6.37	3.15	90	
Switching point	in the MDG							
I-I	12.60	5.73	-	-	-	-	48	
I-AG	11.37	5.99	$10.74^{**}$	5.64	11.29	3.88	93	
I-NAG	12.44	5.60	10.83***	5.33	11.59**	4.24	96	
NAG-I	14.31	5.86	13.73	5.01	14.61	4.46	93	

*Notes:* The switching point reported for I-I are for the first set of decisions; the switching point for the second set of decisions in I-I are 5.79(S.D.=3.60) for the UGand 12.46 (S.D.=5.82) for the MDG; there is no significant difference between the first and the second sets of decisions. The number of group observations differsinthe UG and the MDG because the number of groups reaching unanimity differs in the two games.<sup>\*\*\*</sup> and <sup>\*\*</sup>indicate significance at the 1% and the 5% levels, respectively, in two-tailed Wilcoxon signed rank tests in which the reference is the switching points in the individual decisions.

#### Inequality aversionin individual decisions and in initial proposals in groups

In most treatments, compared to the decisions made in the individual decision-making environment, we find a higher degree of both disadvantageous and advantageous inequality aversion as soon as individuals have to express aninitial choice in a group decision-making environment. Indeed, two-tailed Wilcoxon signed rank tests (W, hereafter)reported in Table 3 show that subjects switch between options significantly later in the UG and sooner in the MDG when making their first proposal in the group compared to the individual decision-making environment.<sup>13</sup>This is observed in all treatments(in I-AG: p=0.006 for the UG and p=0.030 for the MDG;in I-NAG:p<0.001 for both the UG and the MDG; in NAG-I:p=0.009 for the UG), except for the MDG in the NAG-I treatment (p=0.228).This analysis is also largely supported by post-

<sup>&</sup>lt;sup>13</sup>For all thenon-parametric tests reported in this paper, we use individual observations for individual decisions and for initial proposals in the group environment (except that we use group mean values for the comparisons for NAG-I since individual decisions are no longer independent after the group decision-making phase), and we use group observations for the group decisions. Indeed, each initial proposal gives an independent observation since subjects have not interacted yet with the other group members.

regression tests reported in Table A3 in Appendix 3, which are based on random-effects Tobit models reported in Table A2.<sup>14</sup>

We can rule out that this effect is driven by the fact that the initial proposal in groups in I-AG and I-NAG is the subjects' second decision: indeed, the second individual decision in the I-I treatment does not differ from the first one (W tests, p=0.506 in the UG and p=0.942 in the MDG). Moreover, while the switching points for individual decisions are similar in I-I and in the other treatments,<sup>15</sup>the switching points for initial proposals differ from those in the second set of individual decisions in the I-I treatment(in I-AG: p=0.040 for the UG andp=0.068 for the MDG; in I-NAG: p=0.036 for the UG andp=0.080 for the MDG). Note that we find no difference in both UG(p=0.154) and MDG (p=0.266) for the NAG-I treatment. The increased degree of inequality aversion in initial proposals is unlikely driven by the order between the different treatments either. Indeed, comparing the I-NAG and the NAG-I treatments shows no significant difference in the switching points in the UG(M-W tests, p=0.981 for the individual decisions and p=0.633 for the initial proposals). Comparing the I-NAG and the NAG-I treatments shows significant differences in the switching points in the MDG but these differences are observed for boththe individual decisions (p=0.031) and the initial proposals(p<0.001).<sup>16</sup>

Thus, this increased degree of inequality aversion in initial proposals compared to individual decisions is driven by the change from an individual to a group decision-making environment. But which aspect of the group decision-making environment can explain this increase?

<sup>&</sup>lt;sup>14</sup>Table A2 in Appendix 3 reports the marginal effects of variables capturing the decision-making environment and the treatment on the switching points in the UG and the MDG, using Tobit models because of censored data both on the left and on the right (for subjects who never switch). Table A3 reports post-estimation tests.

<sup>&</sup>lt;sup>15</sup>Considering the first individual decisions in I-I, Mann-Whitney tests (M-W, hereafter) give the following *p*-values for the switching points in the UG: 0.672 for I-I *vs.* I-AG, 0.854 for I-I *vs.* I-NAG, and 0.634 for I-AG*vs.* I-NAG. The respective *p*-values in the MDG are 0.277, 0.942, and 0.221. Using instead Fisher's exact tests for categorical outcomes gives the same qualitative conclusions. Kruskal-Wallis tests for I-I *vs.* I-AG*vs.* I-NAG indicate *p*=0.861 for the UG and 0.386 for the MDG.

<sup>&</sup>lt;sup>16</sup>This result does not stem from social information during the aggregation process, as the initial proposals in NAG-I already reveal less inequality aversion than individual decisions in I-NAG.

We can reject that social image drives this finding. Indeed, we find no significant difference in the switching points between the I-AG and I-NAGtreatments as regards the initial proposals (p=0.710 for the UG and p=0.834 for the MDG). We can also reject that this results from efficiency concernsthat more players are affected by the decision when moving from individual to group decision-making environment. Indeed, the difference in switching points is observed in the UG where efficiency, as measured by the sum of payoffs, is kept constant across decisions. Moreover, switching from the selfish to the equal sharing earlier than the eleventh decision problem in the MDG, which is the case for the mean values of the initial proposals n the I-AG and I-NAGtreatments (as shown in the lower panel in Table 3), actually decreases efficiency. The remaining possible explanation is related to the increased number of players participating in the decision. Since subjects received no information about the preferences of their group members, they may have mademore inequality averse proposals either because they believed that the norm of the group was to be more equality concernedor, on the opposite, because they were willingtocounterbalance other players' expectedly more selfish proposals. Since we did not elicit the players' beliefs about others' preferences,<sup>17</sup> we cannot discriminate precisely between these explanations. However, we find that those subjects whose switching points in the initial proposals reveal increased inequality aversion are those whose individual decisions expressed less inequality aversion.<sup>18</sup>An explanation in terms of strategic behavior would require that these more selfish players believe that the others are even more selfish than they are themselves and that they are willing to influence them. Therefore, we believe more in an explanation based on the anticipated social normin the group or on a change of preference due to the higher number of

<sup>&</sup>lt;sup>17</sup>We did not elicit beliefs to avoid hedging and to avoid introducing differences across conditions (precisely, with the individual decision-making environment) that would have affected the payoffs differently across treatments. <sup>18</sup>Spearman coefficients indicate a negative correlation between the values of the switching points in the individual decisions and the difference in switching points between the individual decisions and the individual proposals (p<0.001in both UG and MDG).

players impacted by a group decision. Note that the higher degree of inequality aversion expressed by the initial proposals may have been facilitated by the fact that the initial proposals can be considered as cheap talk if players anticipate that they will not find a unanimous agreement immediately.

Before investigating how this stronger inequality aversion revealed by the switching points in the initial proposals affects the degree of inequality aversion of the final group decisions in Section 4.2, let us compare inequality aversion revealed by the switching points in the individual decisions and in the final group decisions.

#### Inequality aversionin individual decisions and in final group decisions

Table 3 shows that in contrast to the initial proposals, in most treatments the mean switching points for the final group decisions do not differ significantly from the mean switching points in the individual decision-making environment (W tests, p>0.10 in all pair-wise comparisons).<sup>19</sup>The only exception is the I-NAG treatment (p=0.029 for the UGandp=0.051 for the MDG).We need, however, further investigation before concluding that lifting anonymity between group members maintains a higher degree of inequality aversion in groups, sincethe switching points in the final group decisions do not differ between I-AG and I-NAG (M-W tests, p=0.896 for the UG and p=0.584 for the MDG).Finally, comparing the I-NAG and the NAG-I treatments shows no order effect in the degree of disadvantageous inequality aversion in the group decision (M-W, p=0.193), whereas starting a session with group decision-making reduces the degree of advantageous inequality aversion (p<0.001). The dynamics between the initial proposals and the final group decisions is explored in section 4.2.

<sup>&</sup>lt;sup>19</sup>Note that we find no significant differences when comparing the switching points in the second set of individual decisions in I-I and in the group decisions for the MDG in I-AG (M-W, p=0.146) and in I-NAG (p=0.273). However, the differences are significant for the UG in I-AG (p=0.058) and in I-NAG (p=0.041).

#### Econometric analysis

To complement the previous analysis, we now proceed to an econometric analysis. Table 4 reports the marginal effects of Tobit regressions.<sup>20</sup> The dependent variable is the switching point in the UG (columns(1) to (4)) or in the MDG (columns(5) to (8)). Depending on the model, we consider the individual decisions from all treatments in the individual decision-making environment (excluding the second set of decisions in I-I in columns (1) and (5)), or the individual initial proposals and the final group decisions in the group decision-making environment.<sup>21</sup>Moreover, in columns(4) and (8) we report the estimates of random-effects Tobit models afterpooling the data from individual decisions and initial proposals from all treatments.In all regressions, standard errors are clustered at the group level because it is more conservative.

The independent variables include three dummy variables indicating whether or not the MDG was played before the UG, whether or not the session started with the group decisionmaking environment (to control for order effects), and whether or not anonymity was lifted, when appropriate. In the regressions based on the initial proposals (columns (2) and (6)), we also include the switching point from individual decisions. In the regressions based on the finalgroupdecisions (columns (3) and (7)), we include the median switching points, as determined by the three group members' initial proposals. We also include the positive distance (and the negative distance)in initial proposals between the switching point of the member who is above the median (of the player below the median, respectively) in his group and the medianswitching point. This gives an indication of the impact of the skewness of the initial proposaldistribution by different groupmates on the final group decision. We also include in columns (2) and (3) the

<sup>&</sup>lt;sup>20</sup>We use Tobit models because some subjects always choose the same option and never switch between options. Interval regressions (available upon request) provide qualitatively similar results.

<sup>&</sup>lt;sup>21</sup>Note that in models (3) and (7), we consider only one observation per group that achieved unanimity. This is more conservative than taking one observation per group member. To be consistent, in all the other models we only include data from individuals belonging to groups that achieved unanimity.

switching point corresponding to the group offer in the UG because the agreement reached by the group when determining its offer may informplayers about others' preferences, and thus affect the acceptance threshold. In columns (4) and (8), three dummy variables control for individual decisions made first (first set of decisions in I-I and individual decisions in I-AG and I-NAG), for the second set of decisions in I-I, and for individual decisions made after group decisions in NAG-I, with the initial proposals taken as the reference category. Finally, we control for individual characteristicssuch as gender, age, monthly income and number of acquaintances in the same session, except in the models relative to the group decision.<sup>22</sup>

First, Table 4 confirms that, controlling for order effects, subjects express higher disadvantageous and advantageous inequality aversion in their initial proposals in the group decision-making environment than when they decide in isolation: the three individual decision variables all have a negative and significant effect on the switching point in the UG (see columns (4)); they have a positive –albeit not always significant- effect on the switching point in the MDG(see column (8)).<sup>23</sup>Second, columns (3) and (7) show that the median switching points in the initial proposals influence positively the degree of inequality aversion of the final decisions. However, a larger positive distance to the median in the UG (more disadvantageous inequality aversion) and a larger negative distance to the median in the MDG (more advantageous inequality aversion) both have a negative impact on the degree of inequality aversion of the final group decision. This suggests that in the aggregation process the subject who in the group is further from the median player makes more concessions. Third, we find that lifting anonymity

<sup>&</sup>lt;sup>22</sup> Most are not significant. Males express more disadvantageous inequality aversion (column (2), p<0.10). Advantageous inequality aversion is increased by having acquaintances (column (6), p<0.05) and by age (column (8), p<0.001).

<sup>&</sup>lt;sup>23</sup>We also calculated the percentages of individuals who changed their choices across ID, IIP and TD in the UG and the MDG and report the results in Table A4 in Appendix 3. This table indicates that in both UG and MDG: (1) more than half of the subjects change the choices from ID to IIP and from IIP to TD; (2)the percentage of individuals who increase inequality aversion is larger than the percentage decreasing it; (3)the percentages of individuals increasing their degree of inequality aversion from ID to IIP are comparable to the percentage decreasing it from IIP to TD.

within groups does not impact the initial proposals or the final group decision in any game, as the No anonymity variable is never significant. Regarding order effects, we observe that making one's individual decision after the group decision (NAG-I treatment) does not affect significantly any switching point in the UG nor group decisions in MDG, whereas it reduces advantageous inequality aversion in individual decisions and initial proposals in MDG.Finally, we note that when a group has made a more generous offer in the UG, this does not affect the player's initial proposal or the group decision regarding the acceptance threshold (columns(2) and (3)).

Theabove analysis can be summarized as follows.

**Result 1**: Individual initial proposals in groups reveal more inequality aversion than individual decisions made in isolation. Efficiency concerns and social image cannot rationalize this result. Beliefs on the prevailing norm in the group may drive this result.

**Result 2**: However, the degree of both disadvantageous and advantageous inequality aversion is similar in the individual decisions and in the final group decisions.

		Switching p	ointin the UG			Switching po	int in the MD	G
Variables	Individual decision	Initial proposal	Group decision	Indiv. dec and initial proposal	Individual decision	Initial proposal	Group decision	Indiv. dec and initial proposal
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
MDG played before the UG	-1.057**	-0.232	-0.319	-1.030***	-0.373	0.395	-0.538	-0.158
MDO played before the OO	(0.423)	(0.241)	(0.485)	(0.310)	(0.596)	(0.392)	(0.470)	(0.525)
Group offer in the UG		0.135	0.056					
Group oner in the OG		(0.082)	(0.113)					
No anonymity		0.183	-0.004			-0.529	0.546	
		(0.283)	(0.564)			(0.444)	(0.412)	
Individual decision made first (I-AG				-0.874***				0.826***
and I-NAG and 1 <sup>st</sup> set in I-I)				(0.196)				(0.258)
Individual decision made after				-1.081***				0.652
individual decision (2 <sup>nd</sup> set in I-I)				(0.247)				(0.626)
Individual desision made after aroun	-0.030	-0.091	-0.337	-0.669***	2.466***	1.599***	0.181	1.070*
decision (NAG I)	(0.514)	(0.387)	(0.551)	(0.239)	(0.809)	(0.561)	(0.718)	(0.596)
decision (NAC-1)								
SD in individual decision		0.778***				0.677***		
SF III IIIdividual decision		(0.043)				(0.043)		
Group median SPin initial proposals			0.862***				1.051***	
Group median of in mittai proposais			(0.084)				(0.054)	
Dist between above-median and median			-0.214**				0.053	
SP in initial proposals			(0.094)				(0.075)	
Dist between below-median and			0.094				0.184***	
median SP in initial proposals			(0.158)				(0.063)	
Demographics	Yes	Yes	No	Yes	Yes	Yes	No	Yes
Number of observations	318	264	88	636	330	282	94	660
Left-censored observations	38	29	5	72	12	11	0	25
Right-censored observations	1	0	0	1	31	21	5	55
Chi-squared test	0.255	< 0.001	< 0.001	< 0.001	0.022	< 0.001	< 0.001	< 0.001
Log-likelihood	-822.929	-561.318	-187.824	-1484.223	-996.421	-729.888	-208.961	-1875.091

Table 4. Determinants of the switching point in the UG and the MDG (Tobit models)

*Notes:* Marginal effects are reported and standard errors clustered at the group level are in parentheses. In models (4) and (8), standard errorshave been clustered using bootstrapping. MDG played before the UG, no anonymity, individual decisions made first, individual decisions made after individual decisions and individual decisions made after group decisions are dummy variables. "SP" is an abbreviation for "switching point". Demographics include variables capturing gender, age, monthly income and having acquaintances in the session.\*\*\*, \*\*, and \* indicate significance at the 1%, 5% and 10% level, respectively. Including the variable "Group offer in UG" in columns (2) and (3) excludes 2 groups (6 subjects) that did not reach unanimity for proposer decisions in the UG.

#### 4.2. Aggregation of individual choices in groups

To reconcile results 1 and 2, we now explore the aggregation of preferences in groups. We usetwo measures. The first oneis the number of proposal rounds needed to reach unanimity within the group, which captures the tension in the group. The second measure is an individual concession index equal to the mean absolute distance between an individual's initial proposal and the final group decision, divided by the number of rounds needed to converge to a group decision. A higher index means larger concessions per round. We exclude four groups for which the initial proposals were already unanimous and eight groups that did not reach unanimity.

When we pool the three grouptreatments, it takes on average 4.44 rounds (S.D.=3.77) to converge to group decisions on the acceptance threshold in the UG and 4.14 rounds (S.D.=2.11) to converge to the dictators' decisions in the MDG.<sup>24</sup>The number of rounds does not differ significantly between the UG and the MDG (W test, p=0.327). The mean concession index is equal to 0.89 point per round in the UG (S.D.=1.49) and 1.37 (S.D.=2.07) in the MDG.Comparing I-AGand I-NAGreveals no significant differencebased on either the number of rounds or the concession index on the acceptance threshold in the UG (W tests, p=0.849 and 0.909, respectively) or the dictators' decision in MDG(p=0.288 and 0.503, respectively). Finally, comparing I-NAG and NAG-I reveals no order effectbased on the number of rounds orthe concession index in the UG (p=0.408 and 0.727, respectively), as well as on the concession index in the MDG (p=0.589), althoughthe number of roundsdiffers in the latter (p=0.003).

Next, we study who in the group isconverging more rapidly to the group decision. In each group, we rank the subjects based on the median initial proposal and we calculate for each rank the number of rounds until the subject proposes the final group decision. On average, in the UG and the MDG respectively, the median member needs 1.48 and 0.87 fewer rounds to reach the

<sup>&</sup>lt;sup>24</sup> For the group acceptance decisions in UG, the number of rounds is 4.74 in I-AG, 4.19 in I-NAG and 4.44 in NAG-I, and the convergence speed is respectively 1.03, 0.81 and 0.85. For the group dictators' decisions, the number of rounds is 3.68 in I-AG, 4.13 in I-NAG and 4.61 in NAG-I; the convergence speed is respectively 1.51, 1.41 and 1.18.

group decision than the below-median player, and 1.02 and 0.62 fewer rounds than the abovemedian player (M-W tests, p<0.001 in all cases). Considering the mean absolute distance between the initial proposal and the group decision for each rank in the group, we find that it is significantly smaller for the median player (0.40in the UG and 0.05in the MDG) than for the below-median player (0.94 in UG and 0.23 in MDG) and for the above-median player (1.83in the UG and 0.27 in the MDG) (M-W tests, p<0.001 in all comparison tests). This shows that in the UG,concessions are significantly larger for the above-median individuals (who express more disadvantageous inequality aversion than the median player) than for the below-median group members(p<0.001). The difference is not significant in the MDG (p=0.200).Overall, these observations suggest that although each member has a veto power, the aggregation process is driven mainly by the median player. This isconsistent with the results of Ambrus *et al.* (2014).

We next turn to a more formal analysis of the aggregation process. Table 5reports the marginal effects from four regressions in which the dependent variable is either the number of rounds until convergence (Tobit models (1) and (3))<sup>25</sup> or the concession index at the group level (OLS models (2) and (4)). The first two columns are for the group's acceptance threshold in the UG and the last two columns for the dictator group's decisions. The independent variables include the median switching point in the three groupmates' initial proposals, the positive distance between the switching points in the initial proposals for the group member who is above the median and the median, and the corresponding absolute negative distance for the group member who is below the median. They also include three dummy variables indicating whether the MDG was played before the UG, whether the session started with the groupdecision-making environment (equal to 1 for NAG-I and 0 otherwise), and capturing the influence of a lift of anonymity. Standard errors are clustered at the group level.

<sup>&</sup>lt;sup>25</sup>We use Tobit models because the data on the number of rounds needed to converge are censored on the left. Using negative binomial count data models instead of the Tobit models delivers the same qualitative results (available upon request).

	Group acceptance	threshold - UG	Group decision - MDG		
** • • •	Number of	Concession	Number of	Concession	
Variables	proposal rounds	index	proposal rounds	index	
	(1)	(2)	(3)	(4)	
MDG played before the UG	-0.336	0.015	0.027	-0.114	
MDG played before the OG	(0.628)	(0.145)	(0.387)	(0.178)	
Group decision first (NAG I-1)	0.145	0.068	0.277	-0.065	
Gloup decision hist (NAG-I-1)	(0.612)	(0.137)	(0.492)	(0.229)	
Non anonymity	-0.241	-0.154	0.434	0.018	
Non-anonymity	(0.851)	(0.167)	(0.485)	(0.227)	
Group modion SD in initial proposals	-0.139	0.013	0.103*	-0.058**	
Group median SF in mitial proposais	(0.125)	(0.030)	(0.057)	(0.026)	
Distance between above-median and	0.481***	0.101***	0.039	0.145***	
median SP in initial proposals	(0.115)	(0.032)	(0.059)	(0.026)	
Distance between below-median and	0.173	0.143***	0.171***	0.103***	
median SP in initial proposals	(0.169)	(0.044)	(0.053)	(0.031)	
Number of observations	86	86	94	94	
Left censored observations	29	-	16	-	
Chi-squared test	0.013	-	0.009	-	
F-test	-	< 0.001	-	< 0.001	
Pseudo /Adjusted R <sup>2</sup>	0.039	0.307	0.043	0.375	
Log-Likelihood	-188.686	-	-187.421	-	

Table 5. Determinants of the number of proposal rounds and concession index in groups

*Notes:* The regressions include only groups that reached unanimity with at least two rounds of proposals. Models (1) and (3) are Tobit regressions and models (2) and (4) are OLS regressions. Marginal effects are reported and standard errors clustered at the group level are in parentheses. "SP" is the abbreviation for "switching point". \*\*\*, \*\*, and \* indicate significance at the 1%, 5% and 10% level, respectively.

Columns (1) and (2) in Table 5 show that themedian switching point in the initial proposals in the UG has no significant impact on either the speed of convergence or the concession index. But controlling for the median switching point, the higher is the distance above the median (*i.e.*, a higher disadvantageous inequality aversion than the median), significantly higher are both the number of rounds needed for convergence and the concession index per round. And the higher is the distance below the median switching point (*i.e.*, a lower disadvantageous inequality aversion than the median), the higher is the concession index, but not the number of rounds for convergence.

Columns (3) and (4) in Table 5showthat a higher median switching point in the initial proposals in the MDG significantly increases the number of rounds needed to reach unanimity and decreases the concession index per round. Controlling for the median switching point, the greater is the distance below the median(*i.e.*, a higher advantageous inequality aversion than the median), the higher are both the number of rounds and the size of concessions per round. A

greater distance above the median(*i.e.*, a lower advantageous inequality aversion) has no impact on the number of rounds but increases the size of concessions per round.

In both games, the larger is the distance between the median member and the member who is more inequality averse than the median member, the slower the group converges to the final decision: the coefficient for the distance above the median switching point is 0.481 in the UG and the coefficient for the distance below the median is 0.171 in the MDG, both significant at the 1% level. The coefficients are not significant for the distance below the median in the UG and above the median for the MDG. Regarding concessions perround, the larger is the distance both above and below the median switching point, the larger are these concessions: in the UG the respective coefficients are 0.101 (p=0.003) and 0.143 (p=0.002); in MDG, they are 0.145 (p<0.001) and 0.103 (p=0.001). However, the coefficients of the distances below and above the median are not significantly different (t-tests, p=0.403 for the UBG and p=0.269 for the MDG). Finally, Table 5 indicates that the order of games, beginning the experiment with the group decision-making environment, and the lift of anonymitydo not affect the aggregation process. This analysis supports our last result.

**Result 3**: Under the unanimity rule, the aggregation process within groups is driven mainlyby the group member with median preferences.Larger distances both below the median and above the median are associated with largermeanconcessionsper roundand larger concessions in total compared to the initial proposals in the group.A higherdegree of inequality aversion deviating from the median in the group slows down the convergence process.

#### 5. DISCUSSION AND CONCLUSION

Charness and Sutter (2012) statethat groups are "less behavioral than individuals" because they are more likely than individuals to make decisions following standard game-theoretic predictions. Comparing distributional preferences in groups and in individuals, Balafoutas *et al.* (2014) have found that while 15% of individuals can be classified as inequality averse, groupdecision-making eliminates choices consistent with inequality aversion. Our results are somewhat different. First,

we find thatthe degree of disadvantageous and advantageous inequality aversion, asrevealed respectively by the switching points in an ultimatum bargaining game and a modified dictator game, is similar across the groupdecision-making environment and the individual decision-making environment. Second, the initial proposals in groups express more disadvantageous and advantageous inequality aversion than the decisions made in isolation. We suggest that this difference is more likely due to the subjects' beliefs about the prevailing norm in the group or to the change in the environmentrather than to efficiency image concerns. Our design does not allow us to disentangle between these two interpretations; an extension of our work could address directly this question. We observed that when subjects start the experiment with the group decision-making environment their initial proposals show less advantageous inequality aversion than when they start with the individual decision-making environment in itself generates a different initial behavior in the MDG when making an initial proposal to the group. This intriguing observation would deserve additional investigation.

Regarding the convergence process, a third result is that the group decision is mainly influenced by the group member who holds the median preferences. Moreover, astronger degree of inequality aversion above the median revealed by the initial proposals is associated with a slower aggregation process.Finally, anonymity in the group decision-making environment has little overall impact and does not putmore pressure on individuals.This result is surprising since the literature has shown that people whocare about their social imagetend to behave less selfishly when they know they are observed.A possible interpretation is that image matters when the individual decision is the final one, but not when an individual's proposal has to be aggregated with other proposals.Moreover, we have used minimal physical identification: in the noanonymity condition, individuals could see who their group members were, but free-form communication was not allowed. Adding to our design the possibility for the subjects to

exchange verbally, face-to-face, with their group members would reinforce physical identification. This could possibly increase the influence of some individuals with specific social preferences (as shown by Balafoutas *et al.*).

The differences with the results of Balafoutas *et al.* (2014) could result from at least three different sources.First, in their price lists, one option always pays symmetric payoffs whereas in our MDG the fixed option always corresponds to the highest possible inequality. This may have influenced behavior. Second, as already mentioned, verbal deliberations might allow subjects with certain types of preferences to be more assertive than others and affect group thinking. Finally, we have conducted our study in China. In the individual decision-making environment subjects expressed levels of inequality aversion similar to those observed in similar experiments conducted in Europe, despite their exposure to different political and economic institutions. But it would be interesting to further compare the sign and the size of the difference between inequality aversion in individual and groupdecision-making environments in collectivist societies *vs*.individualistic societies.

Other extensions could explore the sensitiveness of inequality aversion in groups to the environmental conditions. Informing the members of newly formed groups about the choices of their groupmates in the individual environment may influence their initial proposals. Replacing simultaneous decision-making with a sequential procedure could also affect the aggregation process. Finally, allowing people to self-select to be part of a group or manipulating the saliency of group identity could also affect the difference in inequality aversion between the individual and the group decision-making environments. This is left for further research.

#### REFERENCES

- Akerlof, G. A., Kranton, R. E. (2000). Economics and identity. *TheQuarterly Journal of Economics*115(3), 715–753.
- Ambrus, A. Greiner, B., Pathak, P. (2014). How individual preferences get aggregated in groups An experimental study. University of New South Wales Discussion Paper.
- Baker, R.J., Laury, S. K., Williams, A. W. (2008). Comparing group and individual behavior in lottery-choice experiments. *Southern Economic Journal* 75, 367–382.
- Balafoutas, L., Kerschbamer, R., Kocher, M., Sutter M. (2014). Revealed distributional preferences: Individuals vs. teams. *Journal of Economic Behavior & Organization*108, 319-330.
- Bartling, B., Fehr, E., Maréchal, M. A., Schunk, D. (2009). Egalitarianism and Competitiveness. *American Economic Review* 99(2), 93-98.
- Bartling, B., von Siemens, F. (2010). Equal Sharing Rules in Partnerships. *Journal of Institutional and Theoretical Economics* 166(2), 299-320.
- Bénabou, R., Tirole, J. (2006). Incentives and Prosocial Behavior. American Economic Review 96(5), 1652-1678.
- Beranek, B., Cubott, R., Gächter, S. (2015). Stated and revealed inequality aversion in three subject pools. *Journal of the Economic Science Association* 1, 43-58.
- Bischoff, I., Krauskopf, T. (2013). Motives of pro-social behavior in individual versus collective decisions a comparative experimental study. Discussion Paper 19-2013, University of Kassel.
- Blanco, M., Engelmann, D., Normann, H. (2011). A Within-Subject Analysis of Other-Regarding Preferences. *Games and Economic Behavior* 72 (2), 321–338.
- Blinder, A.S., Morgan, J.(2005). Are two heads better than one? Monetary policy by committee. *Journal of Money, Credit and Banking* 37, 798–811.
- Bolton, G.E., Ockenfels, A. (2000). ERC: A Theory of Equity, Reciprocity and Competition. *American Economic Review* 90(1), 166–193.
- Bolton, G.E., Ockenfels, A. (2006). Inequality Aversion, Efficiency, and Maximin Preferences in Simple Distribution Experiments: Comment. *American Economic Review* 96(5), 1906-1911.
- Bornstein G., Yaniv I. (1998). Individual and group behavior in the ultimatum game: Are groups more "rational" players? *Experimental Economics* 1(1), 101-108.
- Bornstein, G., Kugler, T., Ziegelmeyer, A. (2004a). Individual and group decisions in the centipede game: Are groups more "rational" players? *Journal of Experimental Social Psychology* 40(5), 599-605.
- Bornstein,G. Schram, A., Sonnemans, J. (2004b). Dodemocracies breed chickens? In Suleiman, R., Budescu, D., Fischer, I.,Messick, D. (Eds.). *Contemporary Psychological Research on Social Dilemmas*. Cambridge University Press, 248–268.
- Bosman, R., Hennig-Schmidt, H., van Winden, F. (2006). Exploring Group Decision Making in a Power-to-Take Experiment. *Experimental Economics* 9, 35–51.

- Brandts, J., Charness, G. (2011). The strategy versus the direct–response method: a first survey of experimental comparisons. *Experimental Economics* 14, 375-398.
- Casari, M., Zhang, J., Jackson, C. (2011). When Do Groups Perform Better than Individuals? A Company Takeover Experiment. Working Paper, University of Bologna.
- Cason, T.N, Mui, V.L. (1997). A Laboratory Study in Group Polarisation in the Team Dictator Game. *Economic Journal* 107(444), 1465-1483.
- Charness, G., Sutter, M. (2012). Groups Make Better Self-Interested Decisions. *Journal of Economic Perspectives* 26(3), 157-176.
- Chen, Y., Li, S. X. (2009). Group identity and social preferences. *American Economic Review* 99(1), 431–457.
- Cheung, S.L., Palan, S. (2011). Two heads are less bubbly than one: team decision-making in an experimental asset market. *Experimental Economics* 15 (3), 373-397.
- Cooper, D.J., Kagel, J.H. (2005). Are two heads better than one? Team versus individual play in signaling games. *American Economic Review* 95, 477-509.
- Cox, J.C. (2002). Trust, reciprocity, and other-regarding preferences: Groups vs. individuals and males vs. females. In Zwick R., Rapoport, A. (Eds.). Advances in Experimental Business Research. Kluwer Academic Publishers.
- Cox, J.C., Hayne, S.C. (2006). Barking up the right tree: Are small groups rational agents? *Experimental Economics* 9(3), 209-222.
- Dannenberg, A., Riechmann, T., Sturm, B., Vogt, C. (2007). Inequality Aversion and Individual Behavior in Public Good Games: An Experimental Investigation. ZEW Discussion Paper 07-034, Mannheim.
- Engelmann, D., Strobel, M. (2004). Inequality Aversion, Efficiency, and Maximin Preferences in Simple Distribution Experiments. *American Economic Review* 94(4), 857–869.
- Fahr, R., Irlenbusch, B. (2011). Who follows the crowd—Groups or individuals? *Journal of Economic Behavior & Organization* 80 (1), 200-209.
- Fehr, E., Schmidt, K. (1999). A Theory of Faimess, Competition, and Cooperation. *The Quarterly Journal of Economics* 114, 817-868.
- Festinger, L., Pepitone, A., Newcomb, T. (1952). Some consequences of deindividuation in a group. *Journal of Social Psychology* 47, 382-389.
- Fischbacher, U. (2007). z-Tree: Zurich Toolbox for Ready-made Economic Experiments. *Experimental Economics* 10(2), 171-178.
- Forsythe, R., Horowitz, J., Savin, N. E., Sefton, M. (1994). Fairness in simple bargaining experiments. *Games and Economic Behavior* 6, 347-369.
- Hoffman, E., McCabe, K., Shachat, K. and Smith, V. (1994). Preferences, property rights and anonymity in bargaining games. *Games and Economic Behavior* 7, 346-380.
- Franzen, A., Pointner, S. (2014). Giving according to preferences: Decision-making in the group dictator game reconsidered. *Soziale Welt* 65(2), 139–156.
- Gangadharan, L., Nikiforakis, N., Villeval, M.C. (2015). Equality concerns and the limits of self-governance in heterogeneous populations. IZA Discussion Paper9384, Bonn.

- Gill, D., Stone, R. (2015). Desert and inequity aversion in teams. *Journal of Public Economics*, 123, 42-54.
- Gillet, J., Schram, A., Sonnemans, J. (2009). The tragedy of the commons revisited: The importance of group decision-making. *Journal of Public Economics* 93, 785-797.
- Güth, W., Schmittberger, R., Schwarze, B. (1982). An Experimental Analysis of Ultimatum Bargaining. *Journal of Economic Behavior &* Organization 3(4), 367–388.
- Güth, W., Levati, M. V., Ploner, M. (2009). Making the World A Better Place: Experimental Evidence from the Generosity Game. Jena Economic Research Paper 2009-071.
- Harrison, G.W., Lau, M.I., Rutström, E.E., Tarazona-Gomez, M. (2012). Preferences over social risk. Oxford Economic Papers 65(1), 25-46.
- He, H., Martinsson, P., Sutter, M. (2012). Group decision making under risk: An experiment with student couples. *Economics Letters* 117, 691–693.
- Kerschbamer, R. (2015). The geometry of distributional preferences and a nonparametric identification approach: The Equality Equivalence Test. *European Economic Review* 76, 85-103.
- Kocher, M., Sutter, M. (2005). The Decision Maker Matters: Individual versus Group Behaviour in Experimental Beauty-Contest Games. *TheEconomic Journal* 115(500), 200–223.
- Kocher, M., Strauß, S., Sutter, M. (2006). Individual or Team Decision-Making Causes and Consequences of Self-Selection. *Games and Economic Behavior* 56(2), 259–270.
- Kocher, M. Sutter, M. (2007). Individual versus Group Behavior and the Role of the Decision Making Procedure in Gift-Exchange Experiments. *Empirica* 34(1), 63-88.
- Kugler, T., Bornstein, G., Kocher, M., Sutter, M. (2007). Trust Between Individuals and Groups: Groups Are Less Trusting Than Individuals But Just as Trustworthy. *Journal of Economic Psychology* 28(6), 646–657.
- Kugler, T., Kausel, E.E., Kocher, M.G. (2012). Are groups more rational than individuals? A review of interactive decision making in groups. *Cognitive Science* 3(4), 471–482.
- Luhan, W., Kocher, M., Sutter, M. (2009). Group Polarization in the Team Dictator Game Reconsidered. *Experimental Economics* 12(1), 26-41.
- Masclet, D., Villeval, M.C. (2008). Punishment, Inequality and Welfare: A Public Good Experiment. *Social Choice and Welfare*, 31(3), 475-502.
- Mohnen, A., Pokorny, K., Sliwka, D. (2008). Transparency, Inequity Aversion, and the Dynamics of Peer Pressure in Teams: Theory and Evidence. *Journal of Labor Economics* 26(4), 693-720.
- Müller, W., Tan, F. (2013). Who acts more like a game theorist? Group and individual play in a sequential market game and the effect of the time horizon. *Games and Economic Behavior* 82, 658–674.
- Nikiforakis, N., Noussair, C.N., Wilkening, T. (2012). Normative Conflict and Feuds: The Limits of Self-Enforcement. *Journal of Public Economics* 96 (9-10), 797-807.
- Reicher, S. D., Spears, R., Postmes, T. (1995). A social identity model of deindividuation phenomena. *European Review of Social Psychology* 6, 161-198.

- Reuben, E., Riedl, A. (2013). Enforcement of Contribution Norms in Public Good Games with Heterogeneous Populations. *Games and Economic Behavior* 77, 122-137.
- Rey-Biel, P. (2008). Inequity aversion and team incentives. *Scandinavian Journal of Economics* 110(2), 297-320.
- Robert C., Carnevale P.J. (1997). Group Choice in Ultimatum Bargaining. *Organizational Behavior and Human Decision Processes* 72(2), 256-279.
- Rockenbach, B., Sadrieh, A., Mathauschek, B. (2007). Teams Take the Better Risks. *Journal of Economic Behavior & Organization* 63(3), 412–422.
- Sheremeta, R.M., Zhang, J. (2010). Can groups solve the problem of over-bidding in contests? *Social Choice and Welfare* 35(2), 175-197.
- Shupp, R.S., Williams, A.W. (2008). Risk preference differentials of small groups and individuals. *TheEconomic Journal* 118(525), 258-283.
- Sniezek, J.A. (1992). Groups under uncertainty: An examination of confidence and decision making by groups. Organizational Behavior and Human Decision Processes 52, 124-155.
- Stahlberg, D., Eller, F., Maass, A., Frey, D. (1995). We knew it all along: Hindsight bias in groups. *Organizational Behavior and Human Decision Processes* 63(1), 46-58.
- Sutter, M. (2005). Are four heads better than two? An experimental beauty-contest game with teams of different size. *Economics Letters* 88(1), 41-46.
- Sutter, M. (2007). Are Teams Prone to Myopic Loss Aversion? An Experimental Study on Individual versus Team Investment Behavior. *Economics Letters* 97(2), 128–132.
- Sutter, M., Kocher, M., Strauß, S. (2009). Individuals and Teams in Auctions. *Oxford Economics Paper* 61(2), 380-394.
- Song, F. (2009). Intergroup trust and reciprocity in strategic interactions: Effects of group decision-making mechanisms. Organizational Behavior and Human Decision Processes 108(1), 164-173.
- Tanaka, T., Camerer, C.F., Nguyen, Q. (2010). Risk and Time Preferences: Linking Experimental and Household Survey Data from Vietnam. *American Economic Review* 100(1), 557–571.
- Van Vugt, M., De Cremer, D., Janssen, D.P. (2007). Gender differences in cooperation and competition: The male-warrior hypothesis. *Psychological Science* 18(1), 19-23.
- Yang, Y., Onderstal, S., Schram, A. (2016). Inequity Aversion Revisited. Journal of Economic Psychology54, 1-16.
- Zhang, J., Casari, M. (2012). How groups reach agreements in risky choices: An experiment. *Economic Inquiry* 50(2), 502-515.

# **Appendix 1. Instructions for the I-NAG treatment (translated from Chinese:** instructions for the other treatments available upon request)

Welcome to this experiment. You have already earned 10 Yuan for showing up on time. During today's experiment, you and the other participants will be asked to make decisions. If you read the following instructions carefully, you can earn a considerable amount of money depending on the decisions you and other participants make. It is therefore important that you take your time to understand the instructions. Please do not communicate with the other participants during the experiment. Should you have any questions, please raise your hand. The experimenters will come to you and answer your question in private.

The experiment consists of several parts. In each part you will be asked to make one or more decisions. You will receive specific instructions before each part begins. The instructions for different parts are different; please read them carefully. Your decisions and answers will remain anonymous unless explicitly specified.

Note that your final earnings from the experiment will be the sum of payoffs from all parts. All payments in the experiment are denoted in points. At the end of the experiment, points will be exchanged to Yuan at a rate of 1 point = 0.03 Yuan.

Your experimental payoff plus the show-up fee will be paid to you in cash in private in another room at the end of the experiment, by an assistant who is not aware of the content of this experiment.

Please do not touch the computer before you are told so, and please do not fold the screen during the entire experiment.

If you have finished reading these instructions and do not have any question, please wait quietly. Otherwise, please raise your hand and the experimenters will come to you and answer your questions in private.

#### Part 1

In this part, there are two roles: Player A and Player B.

Player A is asked to choose between two possible distributions of money between himself/herself and Player B in each of the 21 different decision problems.

Player B knows that A has been asked to make those decisions, and there is nothing s/he can do but accept them.

The role of each participant will be randomly determined as Player A or Player B by the program at the end of the experiment. Which role a participant plays will remain anonymous.

#### **Decisions**

The 21 decision problems will be presented in a chart. Each decision problem will look similar to the following example:

Opti	on X	Opti	Player A's decision		
Player A's Payoff	Player B's Payoff	Player A's Payoff	(Choose	X or Y)	
400	0	100	100	Х	Y

#### You will have to make a decision in the role of Player A.

Hence, if in this particular decision problem you choose Option X, you decide to keep the 400 points for you, so your paired Player B's payoff will be 0 points. Similarly, if you choose Option Y, you and your paired Player B will receive 100 points each.

The 21 rows will be displayed on the computer screens as illustrated in the below chart. The payoffs in Option X are always 400 points for Player A and 0 point for Player B in all decision problems, while the payoffs in Option Y are the same for both Player A and Player B and the payoffs vary from 0 to 400 points in increments of 20 points, in decision problems #1 to #21.

Decision	Option X		Opti	on Y	Dlavor A'	decision
problem #	Player A's	Player B's	Player A's	Player B's	(Choose	A or <b>D</b> )
problem #	Payoff	Payoff	Payoff Payoff		(Choose	$A \text{ or } \mathbf{D}$
1	400	0	0	0	Х	Y
2	400	0	20	20	Х	Y
3	400	0	40	40	Х	Y
4	400	0	60	60	Х	Y
5	400	0	80	80	Х	Y
6	400	0	100	100	Х	Y
7	400	0	120	120	Х	Y
8	400	0	140	140	Х	Y
9	400	0	160	160	Х	Y
10	400	0	180	180	Х	Y
11	400	0	200	200	Х	Y
12	400	0	220	220	Х	Y
13	400	0	240	240	Х	Y
14	400	0	260	260	Х	Y
15	400	0	280	280	Х	Y
16	400	0	300	300	Х	Y
17	400	0	320	320	Х	Y
18	400	0	340	340	Х	Y
19	400	0	360	360	Х	Y
20	400	0	380	380	Х	Y
21	400	0	400	400	Х	Y

The 21 decision problems for Player A (Payoffs in point)

At the end of the experiment, the computer program will randomly assign you as the role of Player A or Player B. If you are assigned the role of Player A, your payoff will be determined as the amount you have chosen for Player A. If you are assigned the role of Player B, your payoff will be determined as the amount your paired participant has chosen for Player B.

You will have to decide the number of the decision problem until which you choose Option X and after which you choose Option Y. You will have to enter an integer between 1 and 21 into one of the two boxes on your computer screen as indicated below, to specify your decision.

#### I choose Option X from decision problem # 1 to decision problem #.

#### I choose Option Y from decision problem # to decision problem # 21 .

Once you enter a number in the range 1-20 in the box in the first line, you must fill in the box in the second line with the number equals to one plus the number in the box in the first line. This means that once you start to choose Option Y in a decision problem, you are not allowed to switch to choose Option X again in any decision problems occurring after this one.

You are also allowed to make the same choice for all 21 decision problems.

If you always choose Option X, you enter the number 21 in the box in the first line. You must keep the box in the second line blank.

If you always choose Option Y, you enter the number 1 in the box in the second line. You must keep the box in the first line blank.

#### Examples

If you enter 21 in the box in the first line, it indicates that you decide to choose Option X in all 21 decision problems.

If you enter 9 in the box in the first line and 10 in the box in the second line, it indicates that you decide to choose Option X from decision problem #1 to decision problem #9 and Option Y from decision problem #10 to decision problem #21.

If you enter 1 in the box in the second line, it indicates that you decide to choose Option Y in all 21 decision problems.

After you have made your choices, please validate your decision by clicking the "Validate" button on your screen.

#### Payoff determination

At the end of the experiment, the computer program will randomly pair you with another participant in the room and will randomly assign the two roles. The computer program will randomly choose one of the 21 decision problems, and the decision outcome in the chosen decision problem will then determine your earnings. The matching and role assignment will remain anonymous. You will make the decision as Player A, but the computer program might assign you the role of Player B when determining payoffs. The assignment of roles is random and does not depend on your decisions as Player A.

If you are assigned the role of Player A, you will receive the amount that you have chosen for Player A in the randomly selected decision problem, and the person paired with you will receive the amount that you have chosen for Player B.

If you are assigned the role of Player B, you will receive the amount that the Player A whom you are paired has chosen for Player B in the randomly selected decision problem.

Before this part begins, a few control questions will be asked to make sure that you have fully understood these instructions. If you have finished reading these instructions and do not have any questions, please wait quietly. The control questions will be displayed on your screen soon. Otherwise, please raise your hand and the experimenterswill come to you and answer your questions in private.

#### Part 2

In this part, there are two roles: Player A and Player B.

Player A is asked to choose one of 21 possible distributions of 400 points between her and Player B.

Player B knows that A has been asked to make this choice, and may either accept the distribution chosen by A or reject it.

If Player B accepts A's proposed distribution, this distribution will be implemented. If B rejects the offer, both receive nothing.

The role of each participant will be randomly determined as Player A or Player B by the program at the end of the experiment. Which role a participant plays will remain anonymous.

#### **Decisions**

The 21 decision problems for Player A and Player B will be presented in a chart. Each decision problem will look similar to the following example:

Distribution cho	Ontion V	Ontion V	Player B's	s decision	
Player A's Payoff	Player B's Payoff	Option A	Option 1	(Choose X or Y)	
300	100	Reject	Accept	Х	Y

#### You will have to make choices in the roles of both Player A and Player B.

If you make choice in the role of Player B, you will have to decide whether you reject or accept each of A's possible 21 proposed distributions. In this example, if you choose Option X, it rejects your paired Player A's proposed distribution and both of your payoffs will be 0 points. If you choose Option Y, A's proposed distribution is accepted; you will receive 100 points and your paired Player A will receive 300 points.

The following chart showing the 21 decision problems will be displayed on your computer screen. The 21 decision problems illustrate the 21 possible distributions of 400 points proposed by Player A, respectively.

	The 21 dee	cision problem	s for Player	B (Payoffs in	point)	
Decision	Distribution Play	proposed by er A	Ontion V	Ontion V	Player B'	s decision
#	Player A's Payoff	Player B's Payoff		Option 1	(Choose	X or Y)
1	400	0	Reject	Accept	Х	Y
2	380	20	Reject	Accept	Х	Y
3	360	40	Reject	Accept	Х	Y
4	340	60	Reject	Accept	Х	Y
5	320	80	Reject	Accept	Х	Y
6	300	100	Reject	Accept	Х	Y
7	280	120	Reject	Accept	Х	Y
8	260	140	Reject	Accept	Х	Y
9	240	160	Reject	Accept	Х	Y
10	220	180	Reject	Accept	Х	Y
11	200	200	Reject	Accept	Х	Y
12	180	220	Reject	Accept	Х	Y
13	160	240	Reject	Accept	Х	Y
14	140	260	Reject	Accept	Х	Y
15	120	280	Reject	Accept	Х	Y
16	100	300	Reject	Accept	Х	Y
17	80	320	Reject	Accept	Х	Y
18	60	340	Reject	Accept	Х	Y
19	40	360	Reject	Accept	Х	Y
20	20	380	Reject	Accept	Х	Y
21	0	400	Reject	Accept	Х	Y

For decision problems #1 to #21, the payoff distributed to Player A reduces from 400 to 0 in increments of 20 points, while the payoff distributed to Player B increases from 0 to 400 in the same increments of 20 points.

In the role of Player A, you will have to choosehow to distribute 400 points payoff between Player A and Player B as stated in one of the 21 decision problems. You will have to enter an integer between 1 and 21 in the box on your computer screen as indicated below, to specify your choice.

### I chooseto distribute the 400 points payoff between me and my paired Player B as the way stated in decision problem # .

**In the role of Player B**, you will have to decide whether you reject or accept each of A's possible 21 proposed distributions. You will have to decide the number of the Player A's proposal until which you reject Player A's proposals (i.e., choose Option X) and after which you accept Player A's proposals (choose Option Y). You will have to enter an integer between 1 and 21 into one of the two boxes on your computer screen as indicated below, to specify your decision.

### I reject the distribution (choose Option X) as shown from decision problem # 1 to decision problem # .

## I accept the distribution (choose Option Y) as shown from decision problem # to decision problem # 21.

Once you enter a number in the range 1-20 in the box in the first line, you must fill in the box in the second line with the number equals to one plus the number in the box in the first line. This means that once you start to accept Player A's proposal in a decision problem, you are not allowed to switch to rejecting the proposals again in any decision problems occurring after this one.

You are also allowed to make the same choice for all 21 decision problems.

If you always reject the proposals of Player A, you enter the number 21 in the box in the first line. You must keep the box in the second line blank.

If you always accept the proposals of Player A, you enter the number 1 in the box in the second line. You must keep the box in the first line blank.

#### Examples

If you enter 21 in the box in the first line, it indicates that you decide to reject Player A's proposals (choose Option X) in all 21 decision problems.

If you enter 9 in the box in the first line and 10 in the box in the second line, it indicates that you decide to reject Player A's proposals (choose Option X) from decision problem #1 to decision problem #9 and accept the proposals (choose Option Y) from decision problem #10 to decision problem #21.

If you enter 1 in the box in the second line, it indicates that you decide to accept Player A's proposals (choose Option Y) in all 21 decision problems.

After you have made your choices, please validate your decisions by clicking the "Validate" button on your screen.

#### Payoff determination

At the end of the experiment, the computer program will randomly pair you with another participant in the room and randomly assign the two roles. The assigned roles and decision outcomes of the two matched participants will then determine your earnings. The matching and the role assignment will remain anonymous.

If you are assigned the role of Player A at the end of the experiment, you will receive the payoff you have chosen for yourself only if your paired person B accepts your offer. Otherwise, both will receive nothing.

If you are assigned the role of Player B at the end of the experiment, you will receive the payoff that your paired Player A has chosen for B, only if you accept that particular offer. Otherwise, both will receive nothing.

Before this part begins, a few control questions will be asked to make sure that you have fully understood these instructions. If you have finished reading these instructions and do not have any question, please wait quietly. The control questions will be displayed on your screen soon. Otherwise, please raise your hand and the experimenters will come to you and answer your questions in private.

#### Part 3

This part is identical to *Part 1*, with one exception. The only difference from Part 1 is that you are now a member of a group, and your group must make group decisions jointly as one decision-maker. Your group consists of three participants in this room.

Please note that your group consists of members with the ID numbers I, II, and III. The other two members in your group are seated next to you in the same row. Members I, II and III are seated at the left, middle and right of the row, respectively. For example, if you are seated at the far right of your row, the two persons to your left from left to right are members I and II, respectively. If you are seated at the far-left of your row, the persons to your left and right are members I and III, respectively. If you are seated at the far-left of your row, the two persons to your right from left to right are members II and III, respectively. If you are seated at the far-left of your row, the two persons to your right from left to right are members II and III, respectively. If you are seated at the far-left of your row, the two persons to your right from left to right are members II and III, respectively. Thus, each member's proposal will be identified by the two other members by his ID number.

In the role of Player A, your group has to make a collective group decision on the number of the decision problem until which you choose Option X and after which you choose Option Y.

Player B makes no decisions.

At the end of the experiment, the computer program will randomly assign your group the role of Player A or the role of Player B.

The three members of the group must make individual proposals and enter them on their computer screens independently. **Unanimity is required for the three membersto reach a collective group decision**. The following procedure determines the group decision:

- The three individual proposals will be simultaneously displayed on all members' screens.
- If the three proposals are not identical, a new proposal round starts. Each member must enter a new proposal. Each member may choose the same proposal as in previous rounds or make a different proposal.
- This group decision-making procedure must be repeated until all group members propose an identical number. This proposal will be automatically converted into the group's decision.
- Members have unlimited number of rounds to enter new proposals in a 10 minute window. Proposals made by each member during previous rounds can be observed in the proposal history box on the right-hand side of the screen.
- If group members have not reached an identical proposal after 10 minutes, the computer program will randomly select one of the possible decisions as the group decision.

Please note that members are not allowed to communicate orally during the entire experiment.

#### Payoff determination

The rules of payoffs determination are identical to that in Part 1.

Please note that each member of the group will receive the determined payoff rather than sharing this amount. That is, for the selected decision, each member in your group will receive this amount.

If you have finished reading these instructions and do not have any questions, please wait quietly. The decision-making screen will be displayed soon. Please enter your proposal as if yourgroup was Player A for this part. Otherwise, please raise your hand and the experimenters will come to you and answer your questions in private.

#### Part 4

This part is identical to *Part 2*, with one exception. The only difference from Part 2 is that now you will be grouped up with the same two other members with the same ID numbers as in *Part 3*, and your group must make groupchoicesjointly as one decision-maker.

In the role of Player A, your group will make a collective groupchoicefor the distribution of 400 points payoff between Player A and Player B as stated in one of the 21 decision problems.

In the role of Player B, your group will make a collective group decision on the number of the Player A's proposal until which you reject Player A's proposals (choose Option X) and after which you accept Player A's proposals (choose Option Y).

At the end of the experiment, the computer program will randomly assign your group the role of Player A or the role of Player B.

The three members of the group must make individual proposals and enter them on their computer screens independently. **Unanimity is required for the three membersto reach a collective group decision**.

The procedure to determine group decisions is identical to that in *Part 3*. In the role of Player A, members have unlimited number of rounds to enter new proposals in a 10 minute window. If group members have not reached an identical proposal after 10 minutes, the computer program will randomly select one of the possible decisions as the groupchoice.

In the role of Player B, the same procedure applies. Group members have again 10 minutes maximum to reach an identical proposal, otherwise the computer program will randomly select one decision as the groupchoice.

#### Payoff determination

The rules of payoffs determination are identical to that in Part 2.

Please note that each member of the group will receive the determined payoff rather than sharing this amount. That is, for the selected decision, each member in your group will receive this amount.

If you have finished reading these instructions and do not have any questions, please wait quietly. The decision-making screen will be displayed soon. Please enter your proposals as if your group was Player A and

Player B, respectively, for this part. Otherwise, please raise your hand and the experimenters will come to you and answer your questions in private.

#### Part 5

You are a member of the same group with the two other members with the same ID numbers as in *Parts 3 and* 4. In this part, your group will participate in a production game.

The production game involves two working groups, Group A and Group B, who are in charge of Departments 1 and 2, respectively. Each group chooses an effort level (an integer between 0 and 100 that is a multiple of 10, i.e., 0, 10, 20, ..., 100), which will determine the production of the department the group is in charge of. A group's total income from this game consists of four parts: (1) Basic salary; (2) A bonus dependent on the production of Department 2; (4) Effort cost, which is dependent on group's own effort level. We introduce each part in turn.

- 1. **Basic salary**. The basic salary is 200 points for Group A and 0 point for Group B.
- 2. **Bonus 1**. The production of Department 1 will be equally divided between Group A and Group B as Bonus 1. Production is wholly determined by Group A's effort level. The higher the effort level Group A chooses, the more Department 1 produces, and, hence, the larger Bonus 1 received by both Group A and Group B.
- 3. **Bonus 2**. The production of Department 2 will be equally divided between Group A and Group B as Bonus 2. Production is wholly determined by Group B's effort level. The higher the effort level Group B chooses, the more Department 2 produces, and, hence, the larger Bonus 2 received by both Group A and Group B.
- 4. **Effort cost**. A group bears the cost of each unit of effort input into the department's production. Each unit of effort in Department 1 costs Group A 2 points. Each unit of effort in Department 2 costs Group B 1 point.

For each group, the total payoff from the production game is represented by the following equation:

#### Total income = Basic salary + Bonus 1+ Bonus 2 - Effort cost.

Please note that, because Group A's basic salary is 200 points while Group B's is 0, total income for Group A is always higher than Group B regardless of the effort levels chosen by Group A and Group B. Of course, the difference varies with different effort levels chosen by the two groups.

After you enter an effort level, you can immediately view the corresponding potential amount of bonus and effort costs displayed. You may test different effort levels to observe the corresponding variation in total income for Group A and Group B. When make your final decisions, ensure that the numbers in the boxes are correct, and press "Submit" at the bottom of the page.

In this part, you will be randomly paired and assigned the role of Group A or Group B. The results of the random pairing and role assignment will remain anonymous and will not be revealed until the end of the experiment. For this reason, every participant is asked to make a decision as Group A and Group B. At the end of the experiment, your decisionfor Group A's effort level will only apply if you are assigned the role of the Group A, otherwise, if you are assigned the role of Group B, your decision for Group B's effort level will adopted.

#### Group decisions

The three members of the group must propose individual proposals and to enter them into their computers independently. **Unanimity is required for the three membersto reach a collective group decision**. Group members must propose individual proposals simultaneously in both the roles of Group A and Group B on the same computer screens. The procedure to determine group decisions is identical to that in *Parts 3 and 4*.

In the roles of Group A and Group B, members have unlimited number of rounds to enter new decisions **in a 20 minute window**.

If group members have not reached identical decisions in the roles of the two types of working groups after 20 minutes, the computer program will randomly select one of the possible decisions as the group decisions for Group A and for Group B, respectively.

#### Payoff determination

Each of the members will receive the determined payoff for a working group rather than sharing this amount. That is, for the selected decision, each of the members in your group will receive this amount.

If you have finished reading these instructions and do not have any questions, please wait quietly. The decision-making screen will be displayed soon. Please enter your proposals as if your group was Group A and Group B, respectively, for this part. Otherwise, please raise your hand and the experimenters will come to you and answer your questions in private.

---

# Appendix 2. Estimation of advantageous and disadvantageous inequality aversion parameters according to Fehr and Schmidt's (1999) model

We can use our data to calculate the parameters of disadvantageous inequality aversion( $\alpha$ ) and advantageous inequality aversion ( $\beta$ ) according to Fehr and Schmidt's (1999) modelin the individual decision-making environment. We start by explaining how to calculate the point estimates, then how to calculate the individual parameters. Finally, we provide statistical tests comparing our data to the estimates from Fehr and Schmidt(1999) and from Blanco *et al.* (2011).

#### 1.Calculation of point estimates

If a subject switches between two points in the UG, this does not mean that he is indifferent at both points; he may be indifferent at one of the endpoints of the interval or at one point in-between. As explained by Blanco *et al.* (2011), to determine a near point estimate of  $\alpha_i$  for each individual, we can suppose that  $s'_i$  is the minimum offer responder *i* is willing to accept and  $s'_i$ - 20 is the highest offer that *i* rejects.

A responder is indifferent between accepting an offer  $s_i \in [s_i' - 20, s_i']$  and rejecting it. Thus,

$$U_i(s_i, 400 - s_i) = s_i - a_i(400 - s_i - s_i) = 0$$
, which gives  $a_i = \frac{s_i}{\left[2(200 - s_i)\right]}$ 

Determining a near point estimate of  $\beta_i$  for each individual requires identifying the decision  $(x_i, x_i)$  for which the dictator in the MDG is indifferent between sharing equally and keeping her 400 points. If she switches to equal sharing at  $(\mathbf{x}'_i, \mathbf{x}'_i)$ , she prefers (400, 0) over  $(\mathbf{x}'_i - 20, \mathbf{x}'_i - 20)$  but  $(\mathbf{x}'_i, \mathbf{x}'_i)$  over (400, 0). Thus, she is indifferent between (400, 0) and  $(\tilde{x}_i, \tilde{x}_i)$ , where  $\tilde{x}_i \in [x_i' - 20, x_i']$  and  $\mathbf{x}'_i \hat{i} \{0, ..., 400\}$ . So,  $\beta_i$  is estimated from the equation  $U_i(400, 0) = U_i(\tilde{x}_i, \tilde{x}_i)$  iff  $400 - 400\beta_i = \tilde{x}_i$ ,

which gives 
$$\beta_i = 1 - \frac{\tilde{x}_i}{400}$$
.

We assume  $\mathbf{S}_i = \mathbf{S}_i - 10$  and  $\tilde{x}_i = x_i' - 10$ . For the responders who accept only  $s_i > 200$  in the UG, we only know that  $\alpha_i \ge 4.5$ , and therefore we consider arbitrarily that  $\alpha_i = 4.5$ , and if  $s'_i = 0$ , we set  $\alpha_i = 0$ .

Similarly, we set  $\beta_i = 0$  for subjects who prefer (400,0) to (400,400) but who perhaps would have  $b_i < 0$ , and we set  $\beta_i = 1$  for subjects who prefer (0,0) over (400,0) but who perhaps would have  $\beta_i > 1$  because we cannot observe a switching point.

# **2.**Calculation of the inequality aversion parameters in the individual decision-making environment

In the individual decision-making environment, individuals compare themselves to a single other individual. Fehr and Schmidt (1999) define utility for n-players as follows:

$$U_{i}(x_{i}, x_{k}) = x_{i} - \partial_{i} \frac{1}{n-1} \mathop{\otimes}\limits_{k=1}^{n} \max(x_{k} - x_{i}, 0) - b_{i} \frac{1}{n-1} \mathop{\otimes}\limits_{k=1}^{n} \max(x_{i} - x_{k}, 0)$$

assuming that  $0 \le \beta_i \le \alpha_i$  and  $\beta_i < 1$ , with  $\alpha$  representing the disadvantageous inequality aversion parameter and  $\beta$  the advantageous inequality aversion parameter, and with  $x_i$  and  $x_k$  representing the payoffs of players *i* and *k*, respectively. In a two-player game, this gives:

$$U(\mathbf{x}_{i},\mathbf{x}_{j}) = \begin{cases} \mathbf{x}_{i} - \partial_{i}(\mathbf{x}_{j} - \mathbf{x}_{i}), & \text{if } \mathbf{x}_{i} \leq \mathbf{x}_{j} \\ \mathbf{x}_{i} - \mathcal{D}_{i}(\mathbf{x}_{i} - \mathbf{x}_{j}), & \text{if } \mathbf{x}_{i} > \mathbf{x}_{j} \end{cases}$$

#### **3.** Distribution of the $\alpha$ and $\beta$ parameters

Table A1 displays the distribution of the  $\alpha$  and  $\beta$  parameters in Fehr and Schmidt (1999), Blanco *et al.* (2011) and in our dataset in the individual decision-making environment, using the same intervals as Fehr and Schmidt (1999) and Blanco *et al.* (2011).

	Disadve	Disadvantageous inequality aversion				Advantageous inequality aversion		
		param	eter (α)		para	parameter $(\hat{\beta})$		
	a.<0.4	0.4≤α	0.92≤α	15/0	β<	0.235≤β	0.5<0	
	α<0.4	< 0.92	<4.5	4.3 <u>≥</u> α	0.235	< 0.5	0.3 <u>≤</u> p	
Fehr and Schmidt (1999)	30%	30%	30%	10%	30%	30%	40%	
Blancoet al. (2011)	31%	33%	23%	13%	29%	15%	56%	
Our data								
Individual environment in all treatments	35%	24%	31%	10%	23%	23%	54%	

Table A1. Distribution of the  $\alpha$  and  $\beta$  parameters of the Fehr and Schmidt's (1999) model in the individual decision-making environment

Using point estimates, Mann-Whitney rank-sum tests<sup>26</sup> indicate no significant difference between the values of  $\alpha$  and  $\beta$  in our experiment and those reported in Blanco *et al.* (2011) (*p*=0.594 for  $\alpha$  and *p*=0.878 for  $\beta$ ).Moreover, the distribution of each parameter is similar in our experiment and in Blanco *et al.* (Kolmogorov-Smirnov tests, *p*=0.234 for  $\alpha$  and *p*=0.562 for  $\beta$ ). In the absence of individual data to compare with Fehr and Schmidt (1999), we conducted Chi-squared tests like Blanco *et al.* with the aggregate data for the distribution percentages in the various categories. There is no significant difference either between our distributions and those reported in Fehr and Schmidt (*p*=0.785 for  $\alpha$  and *p*=0.140 for  $\beta$ ).

Spearman correlation coefficients indicate that  $\alpha$  and  $\beta$  are not correlated in individual decisions made in isolation, in any treatment (*p*>0.10 in all cases). This is consistent with Blanco *et al.* but contrasts with Fehr and Schmidt's assumption. When pooling treatments, we find that 40% of the subjects violate Fehr and Schmidt's assumption that  $\alpha \ge \beta$  when making individual decisions (this percentage was 38% in Blanco *et al.* and 55% in the British sample of Beranek *et al.*, 2015).

<sup>&</sup>lt;sup>26</sup> Considering  $s_i = s'_i - 10$  and  $\tilde{x}_i = x'_i - 10$  in the calculation of the parameters is an approximation that does not impact the results of the non-parametric statistics because they are based on ordinal rankings (see Blanco *et al.*, 2011).

### **Appendix 3. Tables and Figures**

Variables	Switching point in UG	Switching point in MDG		
variables —	(1)	(2)		
Ref.: First individual decision in I-I	-	-		
Second decision in LL	-0.200	-0.139		
Second decision in 1-1	(0.375)	(0.670)		
Individual desision in LAC	0.292	-1.130		
individual decision in I-AG	(0.653)	(0.980)		
	1.112*	-1.730*		
initial proposal in I-AG	(0.651)	(0.979)		
Course desister in LAC	1.722**	-0.797		
Group decision in I-AG	(0.733)	(1.128)		
Individual decision in I-NAG	0.067	0.010		
	(0.637)	(0.976)		
	1.017	-1.603*		
initial proposal in I-NAG	(0.636)	(0.974)		
Course desision in LNAC	1.566**	-1.576		
Group decision in 1-NAG	(0.708)	(1.119)		
Individual desigion in NAC I	0.079	1.838*		
individual decision in NAG-I	(0.643)	(0.978)		
Initial proposal in NAG-I	0.709	1.327		
	(0.643)	(0.979)		
Group decision in NAG-I	0.721	2.656**		
	(0.719)	(1.131)		
Observations	726	754		
Left-censored obs.	77	25		
Right-censored obs.	1	60		
Number of subjects	318	330		
Chi-squared test	< 0.001	< 0.001		
Log-likelihood	-1729.835	-2148.181		

Table A2. Influence of the decision-making environment and of the treatment on switching points in UG and MDG

*Notes:* Regressions are random-effects Tobit models. Reported values are marginal effects. Standard errors are in parentheses. \*\*\* and \*\* indicate significance at the 1% and 5% level, respectively.

Variables	Switching point in UG	Switching point in MDG		
v arrables —	(1)	(2)		
GD vs. ID in I-AG	0.001***	0.650		
GD vs. ID in I-NAG	<0.001***	0.029**		
GD vs. ID in NAG-I	0.127	0.275		
IIP vs. ID in I-AG	0.004***	0.213		
IIP vs. ID in I-NAG	<0.001***	0.001***		
IIP vs. ID in NAG-I	0.021**	0.292		
GD vs. IIP in I-AG	0.166	0.205		
GD vs. IIP in I-NAG	0.179	0.970		
GD vs. IIP in NAG-I	0.976	0.076*		
(GD-ID) in I-AGvs. I-NAG	0.910	0.063*		
(GD-ID) in I-NAGvs. NAG-I	0.144	0.021**		
(IIP-ID) in I-AGvs. I-NAG	0.739	0.135		
(IIP-ID) in I-NAGvs. NAG-I	0.399	0.105		
(GD-IIP) in I-AGvs. I-NAG	0.918	0.381		
(GD-IIP) in I-NAGvs. NAG-I	0.361	0.212		
ID first time vs. second time in I-I	0.594	0.836		
ID in I-I vs. I-AG	0.655	0.249		
ID in I-I vs. I-NAG	0.916	0.992		
ID in I-I vs. NAG-I	0.902	0.060*		
ID in I-AGvs. I-NAG	0.677	0.156		
ID in I-AGvs. NAG-I	0.698	<0.001***		
ID in I-NAGvs. NAG-I	0.982	0.022**		

Table A3.Comparisons between the individual and the groupdecision-making environments based on the estimates of Table A2 (*p*-values from Chi-squared tests)

	Percentage of individuals having different switching points in IIP and ID	Percentage of individuals switching earlier in IIP than in ID	Percentage of individuals switching later in IIP than in ID	Percentage of individuals having different switching points in GD and IIP	Percentage of individuals switching earlier in GD than in IIP	Percentage of individuals switching later in GD than in IIP	Number of subjects
In the UG (accept	ance threshold)						
I-AT	44.05	13.10	30.95**	58.33	35.71	22.62	84
I-NAT	54.17	11.46	42.71***	57.29	31.25	26.04	96
NAT-I	58.89	21.11	37.78**	60.00	34.44	25.56	90
All 3 treatments	52.59	15.19	37.41***	58.52	33.70	24.81*	270
In the MDG							
I-AT	48.39	32.26	16.13**	60.22	24.73	35.48	93
I-NAT	47.92	35.42	12.50***	63.54	25.00	38.54 <sup>*</sup>	96
NAT-I	64.52	32.26	32.26	72.04	34.41	37.63	93
All 3 treatments	53.55	33.33	20.21***	65.25	28.01	37.23 <sup>*</sup>	282

Table A4. Percentage of individuals changing choices across ID, IIP and GD in the UG and the MDG

*Notes:* ID for individual decisions, IIP for individual initial proposals, and GD for groupdecisions. \*\*\*, \*\* and \*indicate significance at the 1%, the 5% and the 10% levels, respectively, in two-tailed Wilcoxon signed rank tests in which the reference iseither the percentage of individuals switching earlier in IIP than in ID or the percentage of individuals switching earlier in GD than in IIP.